Contents lists available at ScienceDirect



journal homepage: www.elsevier.com/locate/iep

Does the internet help the unemployed find jobs?

Eleanor Jawon Choi^{a,b}

^a College of Economics and Finance, Hanyang University, 222 Wangsimni-ro, Seongdong-gu, Seoul 04763, South Korea ^b Institute of Labor Economics (IZA), Schaumburg-Lippe-Strasse 5-9, 53113 Bonn, Germany

ARTICLE INFO

Article history: Received 9 August 2021 Revised 28 October 2022 Accepted 5 January 2023 Available online 9 January 2023

JEL classification: J64 L86 C26

Keywords: Internet job search Unemployment Job-finding rate

1. Introduction

Using the internet for job search and recruitment saw a rapid expansion after the advent of the internet in the mid-1990s. In the US, large commercial job sites, such as Monster.com, Career-Builder.com, and Hotjobs.com, started in 1994–1996. America's Job Bank was founded by the US government in 1995. Craigslist.org, a well-known non-profit advertisement site that includes job posts, was launched in 1995. During the late 1990s and early 2000s, a number of niche job boards and social networking sites, such as LinkedIn, were introduced. The proportion of unemployed workers searching for jobs online increased sixfold from 13% in 1998 to 79% in 2015 (see Fig. 1). Nowadays, the internet provides essential resources and tools for job seekers, and online job search has become even more prevalent with the rise of social media and mobile devices (Cumming et al., 2022; Pew Research Center, 2015; Zhao, 2019). On the employer side, already 94% of the Global 500 companies were recruiting online by 2003, up from only 29% in 1998 (Nakamura et al., 2009).

This study examines the effect of internet job search (IJS) on the employment prospects of unemployed job seekers during the rapid expansion of the internet from the mid-1990s to the early 2010s. While studies demonstrate that the diffusion of the internet substantially improved search efficiency in the product (Brown and Goolsbee, 2002) and marriage (Bellou, 2015) markets, research on

ABSTRACT

This study examines the effect of internet job search (IJS) on job-finding rates among unemployed job seekers during the rapid expansion of the internet from the mid-1990s to the early 2010s. To address endogenous selection into IJS, I use an instrumental variables (IV) strategy exploiting the rise of IJS within occupations over time, which varied across occupations depending on pre-internet exposure to computers at work. The analysis sample includes unemployed workers from the December 1998, August 2000, September 2001, October 2003, and July 2011 Current Population Survey (CPS) Computer and Internet Use Supplements and the September 1992 Basic Monthly CPS, longitudinally matched with their employment outcomes from the subsequent monthly CPS files. The IV estimates indicate that IJS increased the 15-month job-finding rate by 12.9 percentage points (25.1% relative to the mean). Results from placebo exercises and various specification checks support a causal interpretation of the estimated effects. Additionally, the effectiveness of IJS remained stable over time throughout the analysis period.

© 2023 Elsevier B.V. All rights reserved.

internet advantages in job search processes is limited and mixed. Among the few studies focusing on IJS in the US labor market, Kuhn and Skuterud (2004) and Kuhn and Mansour (2014) find that until the early 2000s, unemployment durations among online job seekers were not shorter than their observably similar counterparts who only used offline search methods. Among young workers in the late 2000s, however, Kuhn and Mansour (2014) show that IJS was associated with shorter unemployment spells. I extend the seminal studies of Kuhn and Skuterud (2004) and Kuhn and Mansour (2014) by addressing endogenous selection into IJS and identifying the effectiveness of IJS.

The analysis sample includes unemployed workers aged 16 or above from the December 1998, August 2000, September 2001, October 2003, and July 2011 Current Population Survey (CPS) Computer and Internet Use Supplements and the September 1992 Basic Monthly CPS. Utilizing the CPS survey structure, I longitudinally link individuals from these six baseline surveys with their employment outcomes up to 15 months after, which are observed in subsequent waves of the monthly CPS. Given that the internet was unavailable for general use until 1994, respondents from the September 1992 CPS serve as the pre-intervention sample, in which no one engaged in IJS activities.

As individuals freely decide whether to use the internet for job search, IJS status would be confounded with job seekers' observable and unobservable characteristics. To address this selection problem, I use an instrumental variables (IV) strategy exploiting the rise of IJS within occupations over time, which varied across occupations depending on pre-internet exposure to computers at







E-mail address: choiej@hanyang.ac.kr



Fig. 1. Proportion of Unemployed Workers Searching for Jobs Online.

Notes. Calculations using data from the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements. The 2015 statistic is from Pew Research Center (2015).

work. The instrument is constructed by multiplying a time trend with the occupation-specific computer use rate at work before the emergence of the internet, which is computed using data from the October 1993 CPS School Enrollment Supplement. This empirical strategy is effectively a difference-in-differences IV design in that I exploit variation within occupations over time before and after the intervention in 1994. The instrument strongly predicts differential growth in IJS adoption across occupations.

The key identifying assumption of this IV strategy is that preinternet computer use intensity cannot predict within-occupation changes in job-finding rates without IJS expansion. This assumption would be violated if changes in labor market conditions were systematically correlated with pre-internet computer use intensity across occupations. To address this concern, I perform multiple internal validity checks. First, I conduct a placebo exercise illustrating that the instrument has no predictive power for employment outcomes before internet expansion. Second, I show that the IV estimates are robust to a range of specification checks, especially to the inclusion of control variables representing concurrent labor demand and supply conditions. Lastly, I show that the instrument is unrelated to the probability of switching an occupation, which may exclusively affect the job-finding rate. Results from these exercises suggest that the instrument is unconfounded with preexisting or contemporaneous occupation-specific trends in employment outcomes and thus unlikely to violate the exclusion restriction.

I find that online job seekers are substantially more likely to be employed at various follow-up points than those only searching offline. The IV estimates show that IJS raises the probability of finding a job within the 15-month follow-up period by 12.9 percentage points (25.1% relative to the average employment probability of 0.514). As results from the aforementioned placebo exercise and various specification checks support a causal interpretation of the estimated effects, the IV estimates provide evidence that IJS is effective in facilitating (re-)employment opportunities for the unemployed and reduced search frictions in the labor market. However, the benefits of IJS do not appear in the ordinary least squares (OLS) regression controlling for observable characteristics only. Thus, the difference between the IV and OLS estimates suggests that online job seekers are negatively selected on unobservables. Moreover, I document that IJS effectiveness and selection patterns remained stable over time throughout the analysis period.

This paper contributes to the under-studied literature on online job search and its effectiveness. Kuhn and Skuterud (2004) and Kuhn and Mansour (2014) highlight that online job seekers are positively selected on observables and appear to be negatively selected on unobservables. Only a few studies circumvent this endogeneity problem. Among them, two studies focusing on the expansion of a specific search engine find mixed results. Bagues and Labini (2009) study AlmaLaurea, a job search engine founded in 1994 by a consortium of Italian universities to facilitate the schoolto-work transition of college graduates. Their analysis shows that AlmaLaurea lowered the unemployment rate among graduates of AlmaLaurea member universities. On the other hand, Kroft and Pope (2014) find that the expansion of Craigslist between 2005 and 2007 had no measurable impact on local area unemployment rates in the US. More recently, Bhuller et al. (2021), Denzer et al. (2021), and Gürtzgen et al., 2021 exploit geographic and temporal variation in broadband roll-outs in Germany and Norway. They show that better access to the internet has enhanced both IJS usage and employment prospects among the unemployed. This paper is differentiated from the recent prior works in three aspects. First, the endogeneity of IJS behaviors is addressed at the individual level. Second, I provide more direct evidence of the effectiveness of IJS in the context of the US labor market. Lastly, I examine whether IJS effectiveness and selection patterns have changed over time.

The remainder of this paper is organized as follows. Section 2 describes the construction of the analysis sample and key variables from the CPS data. Section 3 explains the IV estimation strategy used to address the selection problem in the IJS status and measure the effect of IJS on job-finding rates. Section 4 presents the results of IV estimation and internal validity checks. Section 5 discusses the mechanisms and implications of the findings. Section 6 concludes.

2. Data and analysis sample

This section outlines the construction of the analysis sample and key variables. Further details are provided in Sections S1 and S2 of the online appendix.

The main analysis sample includes the unemployed from the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements and the September 1992 Basic Monthly CPS.¹ The internet use supplements surveyed whether the respondents had used the internet to search for work.

To examine employment outcomes, I utilize the CPS rotation group structure and longitudinally match individuals in the 1992/09, 1998/12, 2000/08, 2001/09, 2003/10, and 2011/07 data with the same survey respondents in the ten subsequent Basic Monthly CPS files.² The CPS sample is split into eight rotation groups, each entering the CPS in the same month. Respondents are interviewed for 4 consecutive months, not interviewed for the next 8 months, and re-interviewed for another 4 months before exiting the CPS. Based on this survey structure, each person's employment status can be tracked until a maximum of 15 months after the initial survey.³ Table A1 documents which CPS files are linked from month 0 to months 1-3 and 9-15. For example, respondents from the 2011/07 CPS (i.e., month 0) are merged with their employment outcomes in 2011/08–2011/10 (i.e., months 1–3) and 2012/04-2012/10 (i.e., months 9-15). Those first interviewed in 2011/07 were followed up in the next 3 months until 2011/10, out of the survey for 8 months, and interviewed again for another 4 consecutive months from 2012/07 to 2012/10. Those interviewed in 2011/07 for the fourth time were interviewed again in 2012/04-2012/07, after the 8-month break. Note also that the entire 15month follow-up period of the 1992/09 CPS respondents falls before 1994 (with 1993/12 being month 15). Therefore, their employment outcomes are unaffected by the emergence of the internet.

The analysis sample includes unemployed workers aged 16 or above at baseline surveys in 1992/09, 1998/12, 2000/08, 2001/09, 2003/10, and 2011/07, whose employment outcomes are observed at least once in subsequent monthly surveys.⁴ This yields a sample of 15,655 individuals, 4136 of whom have searched for jobs through the internet.⁵

To construct an instrumental variable exploiting pre-internet exposure to computers at work, I use data from the October 1993 CPS School Enrollment Supplement and calculate the fraction of employed workers using a computer at work by detailed CPS occupation category. Table A2 lists the calculated values of computer use intensity for each occupation in 1993. As the analysis sample is restricted to unemployed workers from the six waves of the CPS, their occupations are classified based on the last job held as of these baseline surveys and linked with the occupation-specific computer use rate calculations. The computer use rate at work cannot be linked to those who have never had a job nor to those who served in the armed forces just before becoming unemployed. Among 15,655 unemployed workers in the analysis sample, 1121 never had a job, and 59 served in the armed forces before the current unemployment spell. For the former, the computer usage rate at work is set to zero.⁶ The average computer use rate across all occupations (i.e., 0.452) is used for the latter.

Table 1 reports summary statistics of variables used in the analysis. Panel A shows the IJS status at the baseline survey (i.e., month 0) and employment outcomes after 1–15 months. Statistics from post-internet (i.e., 1998/12, 2000/08, 2001/09, 2003/10, and 2011/07) data suggest that online job seekers are more likely to be employed in months 9–15 than those searching offline only.

Panel B presents background characteristics measured in the baseline surveys. The unemployed workers who look for work online tend to have observable characteristics associated with a higher likelihood of employment than those who do not engage in online job search activities, as pointed out by Kuhn and Skuterud (2004), Fountain (2005), Stevenson (2009), and Kuhn and Mansour (2014). First, regarding demographic and socioeconomic status, online job seekers are more likely to be of prime working age, better educated, less likely to be Black or Hispanic, and more likely to be married. Second, online job seekers are positively selected on worker characteristics: a higher fraction of online job seekers are voluntary job quitters and have worked prior to the current unemployment spell compared to their offline-only counterparts.⁷ Third, online job seekers are more likely to use each traditional job search method compared to offline searchers.⁸ Lastly, online job seekers are more concentrated in occupations where computer use was traditionally more prevalent, and have better access to the internet at home.

3. Empirical strategy

I estimate the effect of IJS on employment propensities using the following linear probability model:

$$y_i = \alpha + \beta I J S_i + \mathbf{X}'_i \boldsymbol{\gamma} + \delta_k + \lambda_t + \varepsilon_i, \tag{1}$$

⁷ Compared to offline searchers in the same survey year, online job seekers have occupations with lower unemployment rates by 0.8–1.9 percentage points. Pooling all survey years as in Table 1 makes this pattern unclear because online job seekers are disproportionately from more recent data and the occupational unemployment rates are higher in 2011 than in the earlier survey years before the Great Recession.

⁸ The 1992/09 CPS lists six traditional job search methods: 1) contacted employer directly, 2) contacted public employment agency, 3) contacted private employment agency, 4) contacted friends or relatives, 5) placed or answered ads, and 6) other search methods. To ensure that the search method categories are consistent with the 1992/09 file, additional search methods listed in the 1998/12, 2000/08, 2001/09, 2003/10, and 2011/07 CPS are reclassified as other search methods. They include contacted school/university employment center, sent out resumes/filled out application, checked union/professional registers, other active, looked at ads, attended job training programs/courses, and other passive. Note that job search methods categorized in the CPS may or may not involve using the internet.

¹ The analysis is restricted to the civilian noninstitutional population aged 16 or above, which is the universe of labor market statistics published by the Bureau of Labor Statistics (BLS).

² Madrian and Lefgren (2000) describe the CPS longitudinal design and propose approaches to match respondents across CPS surveys.

³ Unlike employment status, each worker's wage cannot be tracked month to month as the CPS collects wage information only among those surveyed for the fourth and eighth times. This makes the analysis of wage outcomes very challenging owing to insufficient observations in the analysis sample.

 $^{^4}$ Under the CPS rotation group system, up to 75% of households at month 0 can be matched to month 1; 50% to months 2 and 12; 37.5% to months 11 and 13; 25% to months 3, 10, and 14; and 12.5% to months 9 and 15. Actual match rates are usually lower than the maximum possible rate due to sample attrition.

⁵ The July 2013 and July 2015 CPS supplements also collected information on IJS status but are excluded from the main analysis for the following reasons. In the 2013/07 supplement, IJS status was not inquired to those in the two rotation groups who were interviewed for the first or the fifth time, making employment outcomes in months 3 and 15 unavailable. In the 2015/07 supplement, IJS status was surveyed only among randomly selected internet users. Hence, using the 2015/07 data, it is infeasible to construct a representative sample of unemployed workers, including both online and offline searchers. Furthermore, occupational categories in both 2013/07 and 2015/07 data are based on the 2010 Census occupational classification system, which is very different from the 1990 Census occupational classification codes

to the 1990 codes is inadequate, creating substantial measurement errors in the instrumental variables. Section S4 in the online appendix provides further details and discusses the results from the extended sample including the 2013/07 and 2015/07 data.

⁶ In Section 4.3, I discuss results from alternative imputation methods or a sample excluding the never-worked.

			Post-inter	net: 1998/12, 2	000/08, 2001/	09, 2003/10, 20	011/07	
	Pre-inter	net: 1992/09	Offline jo	o searchers	Online job searchers			
	Mean (1)	SD (2)	Mean (3)	SD (4)	Mean (5)	SD (6)	Mean diff. (7)	SE (8)
A. Internet job search status and employment	outcomes							
Search for jobs online	0.000	[0.000]	0.000	[0.000]	1.000	[0.000]	1.000	(0.000)
Employed within 15 months	0.504	[0.500]	0.519	[0.500]	0.515	[0.500]	-0.004	(0.012)
Employed in month 1	0.270	[0.444]	0.285	[0.451]	0.225	[0.417]	-0.060	(0.010)**
Employed in month 2	0.330	[0.470]	0.358	[0.479]	0.329	[0.470]	-0.029	(0.014)**
Employed in month 3	0.358	[0.480]	0.386	[0.487]	0.386	[0.487]	0.001	(0.021)
Employed in month 9	0.495	[0.501]	0.489	[0.500]	0.567	[0.496]	0.077	(0.031)**
Employed in month 10	0.502	[0.500]	0.517	[0.500]	0.557	[0.497]	0.040	(0.023)*
Employed in month 11	0.504	[0.500]	0.514	[0.500]	0.570	[0.495]	0.056	(0.019)***
Employed in month 12	0.508	[0.500]	0.504	[0.500]	0.575	[0.495]	0.071	(0.015)***
Employed in month 13	0.527	[0.500]	0.501	[0.500]	0.599	[0.490]	0.098	(0.018)***
Employed in month 14	0.506	[0.500]	0.510	[0.500]	0.642	[0.480]	0.131	(0.021)***
Employed in month 15	0.519	[0.500]	0.520	[0.500]	0.616	[0.487]	0.096	(0.030)***
B. Background characteristics								
Age				10		10	0.05	10
Age 16–25	0.332	[0.471]	0.320	[0.467]	0.229	[0.420]	-0.091	(0.019)***
Age 26–35	0.268	[0.443]	0.202	[0.402]	0.246	[0.431]	0.044	(0.010)***
Age 36–45	0.200	[0.400]	0.207	[0.405]	0.227	[0.419]	0.020	(0.011)*
Age 46–55	0.120	[0.325]	0.153	[0.360]	0.197	[0.397]	0.044	(0.009)***
Education								
Some high school	0.209	[0.407]	0.252	[0.434]	0.089	[0.284]	-0.163	(0.011)***
High school graduate	0.556	[0.497]	0.541	[0.498]	0.496	[0.500]	-0.045	(0.018)**
Some college	0.052	[0.221]	0.047	[0.212]	0.097	[0.296]	0.050	(0.005)***
College graduate	0.132	[0.338]	0.091	[0.287]	0.312	[0.464]	0.221	(0.019)***
Black	0.177	[0.382]	0.192	[0.394]	0.148	[0.356]	-0.044	(0.008)***
Hispanic	0.105	[0.306]	0.167	[0.373]	0.088	[0.284]	-0.079	(0.008)***
Male	0.525	[0.499]	0.509	[0.500]	0.469	[0.499]	-0.039	(0.022)*
Married	0.419	[0.493]	0.362	[0.481]	0.419	[0.493]	0.058	(0.015)***
Married male	0.218	[0.413]	0.175	[0.380]	0.201	[0.401]	0.026	(0.014)*
Spouse employed	0.285	[0.451]	0.294	[0.456]	0.360	[0.480]	0.065	(0.013)***
Home owner	0.568	[0.495]	0.562	[0.496]	0.573	[0.495]	0.011	(0.011)
Duration of unemployment (in weeks)	18.48	[23.39]	17.44	[24.94]	24.69	[29.55]	7.249	(0.963)***
Quit job	0.115	[0.320]	0.106	[0.308]	0.130	[0.336]	0.023	(0.007)***
On layoff	0.113	[0.316]	0.166	[0.372]	0.058	[0.234]	-0.107	(0.011)***
Status before unemployment		[0 = 0 0]		10 1001		10,1001		(0.04.0)
Worked (not including those on layoff)	0.517	[0.500]	0.451	[0.498]	0.634	[0.482]	0.183	(0.016)***
Attended school	0.046	[0.210]	0.163	[0.370]	0.158	[0.365]	-0.005	(0.017)
Did something else	0.324	[0.468]	0.220	[0.414]	0.150	[0.357]	-0.070	(0.009)***
Sector	0 777	[0.410]	0.005	[0.200]	0.010	[0 200]	0.009	(0.022)
Private sector	0.777	[0.416]	0.805	[0.396]	0.812	[0.390]	0.008	(0.023)
Public sector	0.088	[0.283]	0.075	[0.264]	0.108	[0.310]	0.032	(0.011)***
Self-employed	0.028	[0.164]	0.046	[0.209]	0.044	[0.204]	-0.002	(0.005)
State unemployment rate (%)	7.675	[1.378]	5.402	[1.788]	6.485	[2.331]	1.083	(0.148)***
Occupational unemployment rate (%)	7.701	[2.987]	6.320	[3.036]	6.210	[3.321]	-0.110	(0.276)
Traditional job search methods	0.650	[0.476]	0.524	[0.400]	0.500	[0.405]	0.045	(0.012)***
Contacted employer directly	0.652	[0.476]	0.524	[0.499]	0.569	[0.495]	0.045	(0.012)***
Contacted public employment agency	0.196	[0.397]	0.165	[0.371]	0.236	[0.425]	0.072	(0.007)**
Contacted private employment agency	0.081	[0.273]	0.052	[0.222]	0.114	[0.318]	0.062	(0.007)***
Contacted friends or relatives	0.217	[0.413]	0.136	[0.343]	0.206	[0.405]	0.071	(0.009)***
Placed or answered ads	0.369	[0.483]	0.109	[0.312]	0.219	[0.413]	0.110	(0.007)***
Other search methods	0.044	[0.206]	0.547	[0.498]	0.782	[0.413]	0.235	(0.012)**
Number of traditional search methods used Internet access at home	1.559	[1.014]	1.532	[1.102]	2.126	[1.208]	0.594	(0.032)**
	0.000	[0.000]	0.330	[0.470]	0.834	[0.372]	0.504	(0.014)***
Occupational computer use rate in 1993 ^a	0.320	[0.290]	0.283	[0.269]	0.473	[0.301]	0.190	(0.022)***
Ν	3,775		7,744		4,136			

Notes. The baseline characteristics are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements. The employment outcomes are from the subsequent Basic Monthly CPS files matched with the baseline surveys. The number of observations for the employment outcomes varies by follow-up months owing to the rotation group design of the CPS. Standard deviations are reported in brackets. Robust standard errors in parentheses are clustered at the occupation-by-year level. * p < 0.10, ** p < 0.05, *** p < 0.01.

^a The pre-internet computer use at work by occupation is computed using data from the October 1993 CPS School Enrollment Supplement.

where IJS_i indicates whether person *i* searches for work online at the baseline survey (conducted in year t = 1992, 1998, 2000, 2001, 2003, or 2011). The outcome variable y_i is a dummy on whether person *i* becomes employed in a follow-up period, which is 1–15 months after the baseline survey. The regression model includes

fixed effects for person *i*'s occupation $k(\delta_k)$ and baseline survey year $t(\lambda_t)$, which are essential in my IV design that exploits variation within occupations over time. The vector **X**_i controls for person *i*'s demographic characteristics, socioeconomic status, worker characteristics, activities prior to unemployment, and labor market conditions. All of these covariates are observed in the baseline survey.⁹ ε_i is the error term representing the remaining unobserved determinants of the employment outcome.

Because of endogenous selection into IJS, the OLS estimates of the regression model (1) are unlikely to reveal the causal effect of IJS on employment outcomes. To isolate potentially exogenous variation in individuals' IJS status, I employ an IV strategy that exploits within-occupation growth in IJS during the rapid expansion of the internet in the 1990s and 2000s. The instrument projects IJS growth based on pre-internet exposure to computers across occupations. The idea behind this instrument is that individuals with more computer-intensive occupations adopted the internet and IJS earlier and faster than those with less computer-intensive occupations. As the only source of variation in this instrument is the interaction between time and pre-internet computer use intensity across occupations, whether an individual adopted IJS more quickly than others does not affect the projected IJS growth.

IV specification. Based on this idea, I specify my instrument for IJS_i as $C93_k \times \lambda_t$, where $C93_k$ is the proportion of workers using a computer at work in occupation k in 1993 and λ_t represents a set of time dummies. When I present the instrument using indicator functions, the first-stage equation is expressed as follows:

$$IJS_i = \mu + \sum_{l>1994} \theta_l (C93_k \times 1[t=l]) + \mathbf{X}'_i \boldsymbol{\pi} + \delta_k + \lambda_t + u_i,$$
(2)

where 1[t = l] indicates whether the baseline survey year is *l*. As I exploit variation within occupations over time before and after the intervention (i.e., the introduction of the internet to the general public), my empirical strategy is effectively a difference-indifferences IV (DiD-IV) method.¹⁰ In this DiD-IV design, years before 1994 serve as the pre-intervention period. Thus, the dummy for the survey year 1992 is omitted from the regression as a reference year. Standard errors in all regression analyses are clustered at the occupation-by-year level, corresponding to the level of variation in the instrument.¹¹

In Section 4.3, I show that qualitatively similar IV estimates are obtained from alternative specifications for the time trend component in the instrument, such as a linear function (i.e., $C93_k \times t$) or higher order polynomials (i.e., $C93_k \times f(t)$).¹² $C93_k \times \lambda_t$ is used in the main analysis because it employs the most flexible time trend specification and generally yields more conservative estimates than linear or polynomial specifications. Results using $C93_k \times t$ are also reported in the subsequent regression tables for its simplicity and ease of interpretation.

First stage. Figures 2 and A1 plot the expansion of IJS from 1992 to 2011 across occupations with various degrees of pre-internet exposure to computers at work, $C93_k$, illustrating the first-stage relationship. Six selected occupations are presented in Fig. 2 and

¹² When $\text{C93}_k \times t$ or $\text{C93}_k \times f(t)$ is used, my IV strategy becomes similar in spirit to that of Acemoglu and Johnson (2007), where life expectancy is instrumented with predicted mortality constructed as a deterministic function of time and pre-intervention mortality rates from various diseases.

all occupations in Fig. A1. The intensity of pre-internet computer use varies considerably across occupations with $C93_k$ ranging from 0.6% to 98.4%. More importantly, this variation well predicts the differential growth of IJS adoption across occupations. Occupations with a higher fraction of computer users in the pre-internet days exhibit a more rapid increase in the share of online job seekers after the internet became available for general use in 1994. This pattern is depicted in the fanning-out trajectories in Figs. 2 and A1.

For example, before the introduction of the internet, nearly 90% of technicians used computers at work, while less than 1% of private household service workers did so. During the expansion of the internet from pre-1994 to 2011, the fraction of online job seekers grew rapidly to 0.967 among the former and at a much slower rate to 0.318 among the latter. This pattern is more pronounced up to the early 2000s: in 2001, the proportion of unemployed workers searching for jobs online was 0.688 among the former, while staggering at 0.095 among the latter. The average computer use rate at work across all occupations was 45.2% in 1993, close to 42.7% and 49.0% among protective service workers and teachers, respectively. From pre-1994 to 2011, protective service work and the teaching profession saw a steady increase in online job searchers to 0.561 and 0.609, respectively.

Identification assumption. Using the interactions between time and pre-internet computer use intensity as IV for IIS in Eqs. (1) and (2) relies on the assumption that pre-internet exposure to computers at work had no other effect on changes in employment outcomes except through IJS expansion. Note that any level differences across occupations with low and high intensity of initial computer use are captured by occupation fixed effects and thus would not invalidate the identification strategy. The DiD-IV design also allows for any time-varying factors that commonly affect workers in all occupations, such as economic growth or business cycles, which are captured by the survey year fixed effects. Therefore, the only potential threats to the identification strategy are factors that are confounded with initial computer use intensity and exclusively affect within-occupation changes in job-finding rates. To address the concern of violating the exclusion restriction, I conduct multiple internal validity checks in Section 4.2, showing that the instrument is unlikely to be confounded with preexisting or contemporaneous occupation-specific trends in employment outcomes.

4. Results

In this section, I estimate the effect of IJS on the job-finding rate using both OLS and IV methods and examine the internal validity of the empirical strategy. I mainly focus on whether unemployed workers find a job within 15 months and also present whether they are employed in months 1, 2, and 12. The estimated IJS effects for all follow-up months are reported in the online appendix.¹³

4.1. Main results

In Table 2, I first present OLS estimates of the IJS effect on employment probability with and without controlling for covariates X_i . Although controlling for observable characteristics would not sufficiently resolve endogeneity in IJS status, the OLS estimates are presented for comparison with the IV estimates. Panel A shows that the OLS estimates conditional on various control variables are small in magnitude with mixed signs. During the 15-month follow-up period, the difference in the job-finding rate be-

⁹ In addition to occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, selfemployed), industry, and state.

¹⁰ A canonical example using a DiD-IV design is Duflo (2001), who estimates returns to schooling by instrumenting educational attainment with interactions between cohort dummies and intensity of the Indonesian school construction program.

¹¹ Abadie et al. (2017) emphasize that whether or at what level standard errors should be clustered depends on the experimental design. If treatment assignment is made across certain groups, clustering at this group level can be justified in general.

¹³ The IV estimates for follow-up months 9–11 and 13–15 may be subject to a weak instrument bias. See footnote ¹⁷ for further details.



Fig. 2. Proportion of Unemployed Workers Searching for Jobs Online, by Occupation (Selected Occupations). Notes. Online job search status is from the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements. C93 denotes the 1993 computer use rate at work in each occupation from the October 1993 CPS School Enrollment Supplement. Occupation codes follow the 1990 Census occupational classification.

OLS Estimates of Internet Job Search Effects on Employment Probability.

Employed within	Employed	in	
15 months (1)	month 1 (2)	month 2 (3)	month 12 (4)
0.018*	-0.022**	-0.014	0.056***
(0.010)	(0.009)	(0.015)	(0.015)
0.079	0.081	0.083	0.101
rols			
0.027***	-0.026***	-0.005	0.093***
(0.010)	(0.010)	(0.014)	(0.016)
0.022	0.029	0.026	0.028
15,655	13,613	8,679	6,301
0.514	0.265	0.344	0.524
	15 months (1) 0.018* (0.010) 0.079 0ls 0.027*** (0.010) 0.022 15,655	15 month 1 (1) (2) 0.018* -0.022** (0.010) (0.009) 0.079 0.081 ols -0.022** (0.010) (0.010) 0.027*** -0.026*** (0.010) (0.010) 0.022 0.029 15,655 13,613	15 month 1 month 2 (1) (2) (3) 0.018^* -0.022^{**} -0.014 (0.010) (0.009) (0.015) 0.079 0.081 0.083 ols -0.026^{***} -0.005 (0.010) (0.010) (0.014) 0.022 0.029 0.026 $15,655$ $13,613$ $8,679$

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. In panels A and B, occupation and year fixed effects are included in the regressions. In panel B, the regressions also control for unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. * p < 0.10, *** p < 0.05, **** p < 0.01.

tween observably comparable online and offline searchers is 0.018 on average. When individual characteristics are not held constant in panel B, estimated differences tend to be higher than those from controlled OLS regressions. The OLS results confirm that internet job searchers are positively selected on observables as in Kuhn and Skuterud (2004) and Kuhn and Mansour (2014).¹⁴

Table 3 presents IV estimates and describes the main finding of this study: online job seekers are substantially more likely to find a job at various follow-up points than those who only search offline. Panel A reports estimates using $C93_k \times \lambda_t$ as instruments. Overall, IJS raises the employment probability by 12.9 percentage points (with *p*-value = 0.047) during the 15-month follow-up period, which is 25.1% relative to the mean of 0.514.¹⁵ The IV specification using $C93_k \times t$ in panel B tends to provide larger estimates. The estimated effect of searching for jobs online is substantial, considering that average job-finding rates among offline searchers are 28.0%, 34.9%, and 50.5% in months 1, 2, and 12, respectively, and 51.4% during the entire 15-month follow-up period.

In the online appendix, Figure S2 shows that the point estimate for each follow-up month, except months 10–12, is similar to the benchmark estimate for the entire follow-up period. Even the smaller point estimates for months 10–12 are not statistically distinguishable from the benchmark estimate, as standard errors are large. The IJS effects are imprecisely estimated, especially for months 3 and 9–15, owing to smaller sample sizes. As noted in

¹⁴ Based on duration analyses controlling for observable characteristics of job seekers, Kuhn and Skuterud (2004) and Kuhn and Mansour (2014) document that IJS did not reduce unemployment spells during the late 1990s and early 2000s, and Kuhn and Mansour (2014) report stronger benefits among young unemployed workers in the late 2000s. In the online appendix, I compute an OLS version of Kuhn and Skuterud (2004)'s results using CPS data from December 1998 and August 2000. I demonstrate that similar OLS estimates are obtained even when I add the September 1992 CPS data to Kuhn and Skuterud (2004)'s analysis sample and control for year and occupation fixed effects as in Eq. (1) (see Table S1).

¹⁵ In Figure S1, I additionally control for home internet access and/or traditional job search methods, as in Kuhn and Skuterud (2004) and Kuhn and Mansour (2014), and obtain similar estimates to the main specification.

IV	Estimates	of	Internet	Job	Search	Effects	on	Employment	Probability.
----	-----------	----	----------	-----	--------	---------	----	------------	--------------

Dependent variable:	Employed within	Employed in		
	15 months (1)	month 1 (2)	month 2 (3)	month 12 (4)
A. IV: $C93_k \times \lambda_t$				
IJS	0.129**	0.130**	0.105	0.043
	(0.065)	(0.056)	(0.075)	(0.093)
First-stage F-statistic	37.07	32.62	32.86	28.12
B. IV: C93 _k × t				
IJS	0.266**	0.255***	0.177	0.003
	(0.112)	(0.087)	(0.125)	(0.162)
First-stage F-statistic	39.93	35.60	27.67	32.61
Ν	15,655	13,613	8,679	6,301
Mean of Y	0.514	0.265	0.344	0.524

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. The first-stage *F*-statistics are computed by implementing the weak instrument test by Montiel Olea and Pflueger (2013). Robust standard errors in parentheses are clustered at the occupation-by-year level. * p < 0.10, ** p < 0.05, *** p < 0.01.

Section 2, less than 50% of individuals in month 0 can be matched to their employment outcomes after month 2 due to the 4-8-4 rotation group design of the CPS.

The first-stage *F*-statistics reported in Table 3 are computed following Montiel Olea and Pflueger (2013), who provide a weak instrument test robust to heteroskedasticity, autocorrelation, and clustering.¹⁶ They propose rejecting the null hypothesis for weak instruments when the *F*-statistic is greater than 23.1, which is much higher than 10—the conventional rule of thumb established by Stock and Yogo (2005) for cases with homoskedastic and serially uncorrelated errors. As the computed *F*-statistics are above the threshold of 23.1, I can reject the null hypothesis that the instrument, $C93_k \times \lambda_t$ or $C_k^{93} \times t$, is weak.¹⁷ In the online appendix, Table S2 presents the first-stage re-

In the online appendix, Table S2 presents the first-stage relationship between the instrument and IJS.¹⁸ The first-stage estimates using $C_k^{93} \times \lambda_t$ in panel A imply that unemployed individuals who worked in an occupation with a 10 percentage point higher pre-internet computer use rate experienced an additional increase in IJS by 5.59 and 3.36 percentage points from 1992–2001 and 1992–2011, respectively.¹⁹ Those in an occupation with a computer use rate one standard deviation (i.e., 0.294) above would have a 16.4 percentage point higher IJS rate in 2001, which is over onethird of a standard deviation of the 2001 IJS rate (i.e., 45.9%), and a 9.9 percentage point higher IJS rate in 2011, which is approximately one-fifth of a standard deviation of the 2011 IJS rate (i.e., 48.7%). Therefore, the estimated first-stage effect on IJS status is quite large.

4.2. Internal validity

In my IV strategy, a major potential threat to identification is violation of the exclusion restriction. Occupation-specific changes in labor market circumstances may become another channel besides IJS through which the instrument affects job-finding rates. If labor market conditions evolved more favorably for workers in more computer-intensive occupations, IV estimates would be biased upwards. To address this concern, I demonstrate that the instrument is unlikely to be confounded with preexisting or contemporaneous occupation-specific trends in employment outcomes.

Placebo exercises using pre-internet data. I conduct placebo exercises on the reduced-form regression using data from the preinternet period. Before 1994, the internet was unavailable to the general public. Consequently, no first-stage relationship exists between the instrument and IJS, and reduced-form effects should also be zero unless the instrument is confounded with preexisting trends in employment outcomes. For comparison, Table 4 presents reduced-form estimates from the main analysis sample.

In Table 5, the pre-internet data are collected from the September 1983, December 1989, August 1991, and September 1992 Basic Monthly CPS and matched with their subsequent monthly CPS files.²⁰ The placebo sample is designed to mimic the first four waves of the main analysis sample from the 1992/09, 1998/12, 2000/08, and 2001/09 CPS in terms of time coverage and time interval between surveys.²¹ Unlike the positive and significant reduced-form effects during the main analysis period (1992-2011) in Table 4, there is no evidence of such positive relationship between the instrument and employment status in the pre-internet sample (1983-1992). Panel A shows that coefficients on the instruments $C93_k \times \lambda_t$ exhibit no systematic pattern and are not statistically distinguishable from zero either individually or jointly at the 10% significance level. Moreover, in panel B, the coefficient on the instrument $C93_k \times t$ is precisely zero for most employment outcomes or, if statistically significant, negative and very close to zero (e.g., for employment in month 12). These results suggest the lack of preexisting trends in employment outcomes systematically related to pre-internet occupational computer use intensity in 1993.

In the online appendix, I repeat a placebo test on another pre-internet sample from the September 1989, September 1990, September 1991, and September 1992 Basic Monthly CPS, matched with their subsequent monthly CPS files. For this exercise, a shorter time frame is used to focus on labor market conditions during the start of the information revolution in the early 1990s. Moreover, the baseline survey months are fixed to preclude the possibility of occupation-specific seasonality. The reduced-form estimates are reported in Table S3 and are qualitatively similar to those in Table 5. In Figure S3, I also show that employment growth from 1989 to 1993 is unrelated to the level of computer use at work in 1993 across occupations, corroborating the placebo results.

Exploration of contemporaneous occupation-specific trends in employment outcomes. Even in the absence of preexisting trends, labor market conditions may have evolved differently by occupational pre-internet computer usage during the analysis period.

¹⁶ I use the STATA command weakivtest by Pflueger and Wang (2015) to calculate the first-stage *F*-statistic, which is referred to as the effective *F*-statistic in Montiel Olea and Pflueger (2013).

¹⁷ The IV estimates for follow-up months 9–11 and 13–15 in Figure S2 may suffer from a weak instrument problem and need to be interpreted with caution. In particular, $C93_k \times \lambda_t$ becomes a weak instrument with the first-stage *F*-statistic ranging from 8.65 to 19.92 in estimation samples with fewer observations, where employment status in month 9, 10, 11, 13, 14, or 15 is not missing.

 $^{^{18}}$ The coefficient estimates are similar in all columns of Table S2 because observations contributing to the regressions in columns (2)–(4) are subsets of observations used for the regression in column (1).

¹⁹ In panel B using $C_k^{93} \times t$, the first-stage estimate of 0.020 suggests that unemployed individuals in an occupation with a 10 percentage point higher pre-internet computer use rate experienced a 0.2 percentage point additional increase in IJS rate each year.

²⁰ See Table A1 and Section S3 in the online appendix for further details.

²¹ The placebo sample also covers the period of rapid growth in computer use at work and skill-biased technological change in the 1980s (Krueger, 1993; Autor et al., 1998).

Reduced-Form Estimates on Employment Probability.

Dependent variable:	Employed within	Employed in	Employed in			
	15 months (1)	month 1 (2)	month 2 (3)	month 12 (4)		
A. IV: $C93_k \times \lambda_t$						
$C93 \times 1[t = 1998]$	0.120***	0.081*	0.106**	0.163**		
	(0.040)	(0.042)	(0.054)	(0.074)		
$C93 \times 1[t = 2000]$	0.097**	0.085**	0.107**	0.075		
	(0.041)	(0.042)	(0.048)	(0.064)		
$C93 \times 1[t = 2001]$	0.072*	0.058*	0.074	0.017		
	(0.037)	(0.032)	(0.047)	(0.054)		
$C93 \times 1[t = 2003]$	0.048	0.077**	0.023	0.043		
	(0.033)	(0.030)	(0.040)	(0.054)		
$C93 \times 1[t = 2011]$	0.120***	0.100***	0.087*	0.022		
	(0.042)	(0.030)	(0.046)	(0.059)		
F-statistic for joint significance test	2.881	2.647	1.873	1.156		
, ,	[0.015]	[0.023]	[0.099]	[0.331]		
Adjusted R ²	0.079	0.081	0.083	0.100		
B. IV: C93 _k × t						
$C93 \times t$	0.005**	0.005***	0.004	0.000		
	(0.002)	(0.002)	(0.002)	(0.003)		
Adjusted R ²	0.079	0.081	0.083	0.100		
Ν	15,655	13,613	8,679	6,301		
Mean of Y	0.514	0.265	0.344	0.524		

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. *p*-values in brackets. * *p* < 0.10, ** *p* < 0.05, *** *p* < 0.01.

Table 5

Placebo Exercise: Reduced-Form Estimates on Employment Probability, Using CPS 1983-1992.

Dependent variable:	Employed within	Employed in			
	15 months (1)	month 1 (2)	month 2 (3)	month 12 (4)	
A. <i>IV</i> : $C93_k \times \lambda_t$					
$C93 \times 1[t = 1989]$	0.041 (0.031)	0.031 (0.035)	0.040 (0.046)	-0.036 (0.051)	
$C93 \times 1[t = 1991]$	0.012 (0.026)	0.026 (0.033)	0.039 (0.036)	-0.093** (0.039)	
$C93 \times 1[t = 1992]$	-0.031 (0.027)	-0.053 (0.036)	-0.030 (0.043)	-0.059 (0.039)	
F-statistic for joint significance test	2.157	2.802	1.564	2.010	
Adjusted R ²	0.067	0.053	0.062	0.091	
B. <i>IV</i> : C93 _{<i>k</i>} × <i>t</i>					
$C93 \times t$	-0.001 (0.003)	-0.002 (0.004)	-0.000 (0.004)	-0.008** (0.004)	
Adjusted R ²	0.067	0.053	0.062	0.092	
Ν	14,739	12,719	8,108	5,729	
Mean of Y	0.496	0.253	0.330	0.510	

Notes. Data are from the September 1983, December 1989, August 1991, and September 1992 Basic Monthly CPS, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. *p*-values in brackets. * *p* < 0.00, *** *p* < 0.01.

For example, the change in labor demand may have been more dramatic in more computer-intensive occupations during the dotcom bubble and burst between the mid-1990s and early 2000s. On the supply side, workers in occupations with a higher preinternet computer usage may have become more positively selected in terms of skill levels and thus more employable over time due to enhanced technology-skill complementarity. In these situations, my IV strategy may overstate the positive effect of IJS on the job-finding rate.

To address this concern, I first compare the estimated coefficients on IJS across specifications, with and without covariates capturing concurrent labor market conditions or individual char-



Fig. 3. IV Estimates of Internet Job Search Effects on Employment Probability, from Various Specifications.

Notes. Each bullet shape represents the coefficient estimate on *IJS* from a separate IV regression estimating Eq. (1). Vertical spikes around each point estimate represent the 95% confidence interval, clustered at the occupation-by-year level. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. Besides occupation and year fixed effects, the main specification controls for unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state.



Fig. 4. IV Estimates of Internet Job Search Effects on Employment Probability, Controlling for Occupational Employment.

Notes. Each bullet shape represents the coefficient estimate on *IJS* from a separate IV regression estimating Eq. (1). Vertical spikes around each point estimate represent the 95% confidence interval, clustered at the occupation-by-year level. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. Besides occupation and year fixed effects, the main specification controls for unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school, high school, high school, high school, some college college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. The total employment at the occupation group level is from the 1991–1993, 1998, 2000, 2001, May 2003, and May 2011 OEWS.

acteristics. Figure 3 shows that the IV estimates remain stable when state and occupational unemployment rates are dropped from the regression. Without individual covariates, such as demographic characteristics, socioeconomic status, and worker characteristics, the IV estimates change little (in panel A using $C93_k \times \lambda_t$) or tend to be smaller (in panel B using $C93_k \times t$), excluding the possibility of accelerated positive selection of workers in more computer-intensive occupations.

In Fig. 4, I provide further evidence that the instrument is unconfounded with within-occupation changes in labor demand by showing that the IV estimates are robust to additionally controlling for occupational employment. Given the lack of job vacancy data by occupation for the 1990s,²² I use data from the CPS and Occupational Employment and Wage Statistics (OEWS) to compute employment probability and total employment at the occupation

²² For example, the BLS's Job Openings and Labor Turnover Survey (JOLTS) began in 2000 and has been the official source of job openings. The JOLTS data do not provide an occupation breakdown of jobs. The proprietary job postings data constructed by Emsi Burning Glass are available from 2001 onward. The Conference Board has been producing the Help Wanted Online (HWOL) data since 2005.

Reduced-Form Estimates on the Probability of Finding a Job in the Same Occupation.

Dependent variable:	Finding a job in the	same occ. within 15 months
	Main sample (1992–2011) (1)	Placebo sample (1983–1992) (2)
A. IV: $C93_k \times \lambda_t$		
$C93 \times 1[t = 1998]$	0.032	
	(0.051)	
$C93 \times 1[t = 2000]$	-0.067	
	(0.059)	
$C93 \times 1[t = 2001]$	-0.064	
con 41, 20021	(0.051)	
$C93 \times 1[t = 2003]$	-0.046	
CO2 1[4 2011]	(0.057)	
$C93 \times 1[t = 2011]$	0.032	
$C93 \times 1[t = 1989]$	(0.059)	-0.063
		(0.057)
$C93 \times 1[t = 1991]$		0.009
		(0.061)
$C93 \times 1[t = 1992]$		-0.027
		(0.054)
F-statistic for joint significance test	1.472	0.678
_	[0.199]	[0.567]
Adjusted R ²	0.135	0.125
B. IV: C93 _k × t		
$C93 \times t$	0.000	-0.002
	(0.003)	(0.006)
Adjusted R ²	0.135	0.125
Ν	7,552	6,520
Mean of Y	0.393	0.409

Notes. In column (1), data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. In column (2), data are from the September 1983, December 1989, August 1991, and September 1992 Basic Monthly CPS, matched with their subsequent monthly CPS files. In both columns (1) and (2), the estimation is conducted on those who have found a job during the 15-month follow-up period and whose occupation is not missing both in month 0 and after becoming employed. C93 denotes the 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on lay-off, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. *p*-values in brackets. * p < 0.10, ** p < 0.05, *** p < 0.01.

level as a proxy for occupational labor market tightness.^{23,24} The estimated coefficient on IJS is insensitive to adding one of the occupational employment measures in the IV regression.^{25,26}

Lastly, I show that pre-internet computer use intensity is unrelated to over-time changes in the probability of finding a job within the same occupation. One concern motivating this exercise is that for those who worked in an occupation with higher exposure to computers, it may have become easier to find a job by switching to another occupation. If easier job switching and, consequently, a higher job-finding rate are attributable to increased demand for computer-accustomed workers rather than to IJS growth, this is a possible scenario in which the instrument might directly affect the employment outcome. Table 6 reports the estimated reduced-form effects on the probability of finding a job within the same detailed occupational category for the analysis period (1992-2011) in column (1), in comparison with the estimates for the placebo period (1983–1992) in column (2).²⁷ In both columns, the coefficients on the instruments are statistically indistinguishable from zero at any conventional level of significance for both IV specifications, $C93_k \times \lambda_t$ and $C93_k \times t$.²⁸ This provides another piece of evidence against violation of the exclusion restriction during the analysis period.

²³ Both CPS and OEWS have their own caveats in counting the number of jobs. As the CPS is "person-based" and focuses on the main job of respondents, many of the secondary jobs are not considered. The OEWS is "job-based" but does not cover unincorporated self-employed, agriculture, or private household jobs. See Abraham and Spletzer (2010) for a comparison of occupational employment between CPS and OEWS.

²⁴ It is very difficult to harmonize the occupation codes used by OEWS and CPS/Census, and no crosswalk is available especially for the 1990s. To reduce measurement errors from manual reclassification, I convert the OEWS occupation codes to 14 broad occupation groups instead of 45 detailed occupation categories in the 1990 Census occupational classification system. Section S4 in the online appendix provides further details on how occupational employment is constructed using the OEWS data.

 $^{^{\}rm 25}$ Controlling for occupational employment in logs instead of levels also gives similar IV estimates.

²⁶ The missing values of the occupational employment measures among veterans or the never-worked are imputed with the sample averages across all occupations.

 $^{^{\}rm 27}$ The estimation is conducted on unemployed workers 1) who have found a job during the 15-month follow-up period and 2) whose occupation is not missing both in month 0 and after becoming employed.

 $^{^{\ 28}}$ This pattern is confirmed by an occupation-level analysis presented in Figure S4.

The results of the aforementioned internal validity checks suggest that the positive IJS effect estimates from the proposed IV method are unlikely to be driven by differential changes in labor market conditions across occupations.

4.3. Robustness checks

In this section, I present IV estimates from alternative specifications for the instrument and from interval regressions that consider censored unemployment spells. I also discuss whether and how the results change when the pre-internet computer use rate at work among the never-worked is handled differently instead of being imputed with zero.

IV specifications. Table A3 compares IV estimates from alternative IV specifications: $C93_k \times t$, $C93_k \times f(t)$, and $C93_k \times \lambda_t$. When I impose a functional form assumption on the time component of the instrument using a linear, quadratic, or cubic time trend, IV estimates remain qualitatively similar but become larger in magnitude compared to the preferred specification using a set of time dummies. In most cases, the preferred specification, $C93_k \times \lambda_t$, provides the most conservative IJS effect estimates, and the linear specification, $C93_k \times t$, returns the largest estimates.

Censored regression. For the main analysis, I have estimated the linear probability model in Eq. (1) instead of a duration model because dealing with endogenous regressors is easier when both the outcome (i.e., employment status) and treatment (i.e., IJS status) processes are linear. When the outcome and treatment variables originate from nonlinear and nonnormal processes, particularly from different statistical families, tractable multivariate distributions for a joint model of outcome and treatment often do not exist (Cameron and Trivedi, 2005). Duration outcomes with binomial treatment are a typical example with no analytic joint distribution.

Nevertheless, the linear probability analysis of employment status observed within a certain period or in a specific follow-up month has limitations. Notably, the linear probability model used in the main analysis does not capture that unemployment spells are censored in the CPS: 1) unemployment may have continued after individuals stopped being followed up by the CPS, and 2) individuals may have been employed during the 8-month gap between the two 4-month CPS survey windows. To incorporate these features from the CPS survey structure, I examine the timing of employment using a censored regression model in Table 7. Column (1) shows that online job seekers stay unemployed slightly longer than their observably similar counterparts who do not search online, suggesting negative selection into IJS on unobservables. In columns (2) and (3), the outcome and first-stage equations are jointly estimated using the maximum likelihood estimation to address selection into IJS, following the literature on limited dependent variable models with endogenous regressors (e.g., Angrist, 2001). Results suggest that IJS accelerated the timing of employment by 0.6-1.3 months (13.7-32.3% relative to the mean of 4.7 months until employment), which is consistent with the main findings from the IV analysis using the linear probability model.

Alternative treatments of the never-worked. For those who never had a job, the computer use rate at work in 1993, and thus the instrument, is missing and imputed with zero in the main analysis. Table 8 presents the IV estimates of the IJS effect on the 15-month job-finding rate using alternative imputed values of C93 for the never-worked or from an alternative estimation sample excluding the never-worked. The reduced-form estimates are reported in Table A4. In columns (2)–(4), I impute C93 with 0.053, 0.108, and 0.175, respectively, instead of zero, for those who never had a job. These values are the pre-internet computer use rate at work

Table 7

Censored Regression Estimates of Internet Job Search Effects on Unemployment Du-
ration since Month 0.

	Censored reg.	Censored reg. with IV		
	(1)		$\begin{array}{c} C93_k \times t \\ (3) \end{array}$	
IJS	0.074**	-0.646**	-1.329***	
	(0.037)	(0.254)	(0.436)	
Log likelihood	-13,542	-19,565	-19,668	
N	15,655	15,655	15,655	
Mean of Y	4.682	4.682	4.682	
Median of Y	2.0	2.0	2.0	

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation by-year level. * p < 0.10, ** p < 0.05, *** p < 0.01.

in occupations 31, 29, and 36, which display over-time trajectories of the IJS rate close to the never-worked group (see Fig. A1). The IV and reduced-form estimates remain similar when the nonzero values are used for imputation. In column (5), the restricted sample without the never-worked tends to yield smaller IV and reduced-form estimates, suggesting that IJS is more effective for those who never had a job before. A possible explanation is that IJS may be more beneficial to job seekers who do not have career networks and thus have limited access to insider information on job opportunities (Cumming et al., 2022).

5. Discussion

This section discusses the underlying mechanisms and implications of the findings in relation to the existing literature.

Theoretical framework. Online job search and recruitment tools improve match efficiency and reduce search costs by facilitating information exchanges between workers and firms.²⁹ Specifically, online job boards make learning about and applying for jobs easier and cheaper. Internet technology also enhances a firm's ability to announce vacancies and reach job candidates.

Labor market search theory suggests that greater search efficiency and lower search costs can have both positive and negative effects on the employment probability among job seekers (e.g., Mortensen and Pissarides, 1999; Pissarides, 2000). More potential matches between firms and workers are expected to create more hires and increase job-finding rates among the unemployed.³⁰ However, workers may become more selective regarding the quality of matches when considering more, faster, and cheaper potential matches. The consequent increase in reservation wages will then lower the job-finding rate.

Concerning the strength of the two competing effects, Gürtzgen et al., 2021 predict that internet expansion would most likely raise the job-finding rate, referring to

 $^{^{29}}$ Autor (2001), Bhuller et al. (2021), and Gürtzgen et al., 2021 provide an overview of the expected impacts of internet expansion on various labor market outcomes.

³⁰ In addition, Bhuller et al. (2021) point out that reduced recruitment costs are equivalent to enhanced productivity for firms and would increase the number of job vacancies per job seeker, leading to a higher job-finding rate.

Г	ble	8	

IV Estimates of Internet Job Search Effects on Employment Probability, by Alternative Treatments of the Never-Worked.

Dependent variable:	Employed wi	thin 15 months			
	Main sample (1)	Impute with C93 = 0.053 (2)	Impute with C93 = 0.108 (3)	Impute with C93 = 0.175 (4)	Without the never-worked (5)
A. IV: $C93_k \times \lambda_t$					
IJS	0.129**	0.119*	0.108*	0.093	0.056
	(0.065)	(0.062)	(0.060)	(0.057)	(0.054)
First-stage F-statistic	37.07	37.06	36.73	36.26	35.07
B. IV: C93 _k × t					
IJS	0.266**	0.254**	0.242**	0.226**	0.184*
-	(0.112)	(0.110)	(0.109)	(0.107)	(0.108)
First-stage F-statistic	39.93	41.24	42.10	42.59	42.72
Ν	15,655	15,655	15,655	15,655	14,534
Mean of Y	0.514	0.514	0.514	0.514	0.522

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Column (1) reproduces the main IV estimates in Table 3. In columns (2)–(4), C93 for those without work experience is imputed with the value of C93 in occupation categories 31, 29, and 36, respectively. In column (5), the estimation sample excludes individuals who never worked before the current unemployment spell. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. The first-stage *F*-statistics are computed by implementing the weak instrument test by Montiel Olea and Pflueger (2013). Robust standard errors in parentheses are clustered at the occupation-by-year level. * p < 0.10, *** p < 0.05, *** p < 0.01.

Van den Berg (1994)'s theoretical work. Regarding the impact of increased job offer arrival rates on the employment probability, Van den Berg (1994) shows that the positive effect from increased employment opportunities dominates the negative effect through increased reservation wages and prolonged job search for virtually all wage offer distributions used in the literature.

Interpretation of IV results. The IV results in Section 4.1 indicate that IJS substantially improved search effectiveness among unemployed individuals in the US labor market during the internet expansion from the mid-1990s to the early 2010s. This is in line with recent empirical evidence from Germany and Norway that access to broadband internet generated more employment among job seekers and reduced search frictions in the labor market (Bhuller et al., 2021; Denzer et al., 2021; Gürtzgen et al., 2021).

My IV estimates cannot be directly compared to the estimates from these studies since they focus on the effect of internet availability on employment prospects and provide only suggestive evidence of the effectiveness of online job search. Nevertheless, I can conduct a back-of-the-envelope calculation using the estimates from Bhuller et al. (2021), who measure the effects of internet access on the probability of finding a job within 2 years (0.016) and on the likelihood of using the internet to browse job advertisements (0.088). An IV estimate of 0.182 (= 0.016/0.088) can be obtained when the former is regarded as the reduced-form estimate and the latter as the first-stage estimate, treating broadband availability as the instrument.³¹ Despite the difference in the population of interest and variation exploited, the back-of-the-envelope calculation result is within the range of my IV estimates between 0.129 and 0.266 for the 15-month job-finding rate in Table 3.

Bias in OLS estimates. A comparison between the OLS and IV results reveals that the OLS estimates are downward biased, suggesting negative selection into IJS on unobserved characteristics.³² The following are two possible reasons for negative selection. First, the

drastically reduced cost of job searches on the internet may have widened the pool of job seekers to nonserious or less qualified applicants (Autor, 2001). Because reading online ads, searching for job listings on the web, or posting a resume on online job boards is almost free, even individuals with low expected benefits (i.e., low employment chances) may become willing to engage in such online job search activities.

Second, online job seekers may have poor social networks and informal contacts compared to their observably similar counterparts who only search offline. Specifically, job seekers with a weaker social network mainly resort to formal search channels, and online job search had remained disproportionately formal until 2008 (Gürtzgen et al., 2021; Kuhn and Mansour, 2014).³³ If the benefits of informal channels are concentrated in early unemployment stages and depleted relatively quickly (Van den Berg and Van der Klaauw, 2006), the downward bias in OLS estimates would be especially severe at the beginning of the 15-month follow-up period. Indeed, the OLS estimates for month 9 or later are not only more distinctively positive and higher but also closer to the benchmark IV estimate (i.e., 0.129 from $C93_k \times \lambda_t$ and 0.266 from $C93_k \times t$), compared to the OLS estimates for months 1 and 2 (see Table 2 panel A, Table 3, and Figure S2).

IJS effects over time and across subgroups. Given the dramatic increase in IJS from the mid-1990s to the early 2010s, the effectiveness and selection pattern of IJS may have changed over time. In Table 9, I explore this possibility by examining whether the IV and OLS estimates of the IJS effects in more recent years (2011) differ from the early expansion period (1998–2003). Although the IV estimates tend to be larger in 2011 than in 1998–2003, the IJS effects in the two periods are not statistically distinguishable.³⁴ In addition, the OLS estimates are similar across years and remain

³¹ This estimate implies a 27.6% increase relative to the average employment probability of 0.659 in the Norwegian sample analyzed by Bhuller et al. (2021).

 $^{^{32}}$ Gürtzgen et al., 2021 point out that male and young online job seekers especially appear to be negatively selected.

³³ According to Kuhn and Mansour (2014)'s calculation using the 2008 wave of the National Longitudinal Survey of Youth 1997 (NLSY97), among the 12 active and passive job search methods, "sending out resumes or applications" and "looking at advertisement" were the most internet-intensive, while "contacting friends or relatives" was the least internet-intensive.

 $^{^{34}}$ As shown in Figure S5, the 95% confidence interval for the IJS effect in 2011 overlaps with that in 1998–2003 for all the follow-up months.

Heterogeneous Effects of Internet Job Search on Employment Probability, by Year Group.

Dependent variable:	Employed within	Employed in	Employed in		
	15 months	month 1	month 2	month 12	
	(1)	(2)	(3)	(4)	
A. IV: $C93_k \times \lambda_t$					
$IJS \times 1[t = 1998, 2000, 2001, 2003]$	0.123*	0.126**	0.106	0.044	
	(0.065)	(0.055)	(0.075)	(0.093)	
$IJS \times 1[t = 2011]$	0.296**	0.260***	0.209	-0.032	
	(0.127)	(0.095)	(0.143)	(0.170)	
χ^2 -statistic testing IJS effects	2.484	2.993	0.756	0.300	
identical across year groups	[0.115]	[0.084]	[0.385]	[0.584]	
B. <i>IV</i> : C93 _{<i>k</i>} × <i>t</i>					
$I/S \times 1[t = 1998, 2000, 2001, 2003]$	0.185**	0.171***	0.156**	0.166	
	(0.072)	(0.063)	(0.079)	(0.117)	
$I_{s} \times 1[t = 2011]$	0.352***	0.300***	0.260*	0.078	
	(0.130)	(0.098)	(0.146)	(0.179)	
χ^2 -statistic testing IJS effects	2.254	2.671	0.730	0.385	
identical across year groups	[0.133]	[0.102]	[0.393]	[0.535]	
C. OLS					
$I_{IS} \times 1[t = 1998, 2000, 2001, 2003]$	0.020*	-0.019*	-0.007	0.051***	
	(0.012)	(0.011)	(0.017)	(0.018)	
$I_{IS} \times 1[t = 2011]$	0.010	-0.032**	-0.036	0.069***	
5 1 1	(0.018)	(0.015)	(0.026)	(0.023)	
F-statistic testing IJS effects	0.255	0.599	0.922	0.353	
identical across year groups	[0.614]	[0.440]	[0.338]	[0.553]	
Adjusted R ²	0.079	0.081	0.083	0.101	
Ν	15,655	13,613	8,679	6,301	
Mean of Y	0.514	0.265	0.344	0.524	

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. *p*-values in brackets. * *p* < .10, ** *p* < .05, *** *p* < .01



Fig. 5. Heterogeneous Effects of Internet Job Search on Employment Probability, by Various Subgroups.

Notes. A set of the same bullet shapes represent the coefficient estimates on the interaction terms between *IJS* and group dummies from an IV regression estimating Eq. (3). Different types of bullet shapes are from separate regressions. Vertical spikes around each point estimate represent the 95% confidence interval, clustered at the occupation-by-year level. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. In panel B, the 95% confidence interval of the IJS effect for those with some high school education or less is [-2.181, 2.287], which is beyond the range of the horizontal axis.

substantially smaller than the IV estimates. This suggests no significant change in the selection pattern. Therefore, IJS has constantly been effective in securing jobs for one and a half decades since the introduction of the internet.

Additionally, I examine whether the benefits of IJS are heterogeneous by demographic characteristics, education level, or retrospective unemployment duration. Fig. 5 presents estimated coefficients (i.e., β_g 's) on the interaction terms between the IJS status and subgroup dummies in the following regression specification:

$$y_i = \alpha + \sum_g \beta_g I J S_i \times \mathbf{1}[G_i = g] + \mathbf{X}_i' \boldsymbol{\gamma} + \delta_k + \lambda_t + \varepsilon_i, \tag{3}$$

where $1[G_i = g]$ indicates whether person *i* belongs to group *g*. For example, I include two interaction terms between the IJS status and each sex dummy in the regression to allow IJS effects to differ between men and women. I find no evidence of heterogeneous IJS effects by sex, age, education, or long-term unemployment status.^{35,36} Although less precisely estimated, subgroup-specific IJS effects are generally similar in magnitude to the average effect for all unemployed workers in Table 3. Also, I cannot reject the null hypothesis that the IJS effects are identical across subgroups at the 10% significance level.

6. Conclusion

This study investigates whether using the internet for job search helps unemployed workers find jobs. To address the selfselection of online job seekers, I employ an IV strategy based on the fact that IJS was adopted more rapidly among individuals with occupations that had greater exposure to computers at work prior to the emergence of the internet. The results of my analysis of unemployed workers from the 1992/09, 1998/12, 2000/08, 2001/09, 2003/10, and 2011/07 CPS indicate that online job seekers are substantially more likely to be employed during the 15-month followup period than those who search offline only. Therefore, IJS effectively raised job-finding rates among the unemployed during the internet expansion from the mid-1990s to the early 2010s. The IV results also suggest that, between the two competing channels, the positive effect of improved search efficiency dominates the negative effect of increased reservation match quality and prolonged job search. The effectiveness of IJS remained stable throughout the analysis period.

The following are two limitations of this study. First, the IV analysis of individual outcomes identifies the partial equilibrium effect of IJS on the job-finding rate. To address the general equilibrium question of whether improved employment prospects for online job seekers lead to a reduction in aggregate unemployment, search externalities on non-internet searchers should also be con-

sidered at the labor market level, which is beyond the scope of this paper. Second, this study focuses on (re-)employment outcomes among the unemployed mainly due to data limitations. Whether the internet advantage can be extended to other outcomes (e.g., wages) or different groups of workers (e.g., the employed) remains an important subject for future research.

Declaration of competing interest

I acknowledge research funding from the Korea Sanhak Foundation and Hanyang University in Seoul, South Korea. The funding source had no involvement in the conduct of the research or preparation of the manuscript. I have no competing financial interests or personal relationships that have influenced the work reported in this paper. IRB approval was not needed for the project, as all data are non-identifiable.

CRediT authorship contribution statement

Eleanor Jawon Choi: Conceptualization, Data curation, Formal analysis, Funding acquisition, Investigation, Methodology, Project administration, Resources, Software, Supervision, Validation, Visualization, Writing – original draft, Writing – review & editing.

Data availability

The data used in my analyses are from the Current Population Survey (CPS) and the Occupational Employment and Wage Statistics (OEWS) in the United States. The CPS data are publicly available from the National Bureau of Economic Research: https://www. nber.org/research/data/current-population-survey-cps-data-nber. The IPUMS CPS harmonizes a subset of variables in the raw CPS files to facilitate cross-time comparisons and provides the integrated datasets at https://cps.ipums.org/cps/. The OEWS data are publicly available from the Bureau of Labor Statistics: https://www.bls.gov/oes/tables.htm.

Acknowledgments

I thank the editor, two anonymous referees, Hank Farber, Elizabeth Weber Handwerker, Peter Kuhn, Yan Lau, David Lee, Johanna Catherine Maclean, Anne Polivka, Jim Spletzer, and conference participants at AMES, APPAM, and SEA for helpful comments and discussions. I also thank Elizabeth Kayoon Hur, Hanbin Kim, Jung Hyun Kim, and Young Joon Kim for excellent research assistance. This work was supported by the Korea Sanhak Foundation in 2020 and the research fund of Hanyang University (HY-20210000003530). All errors are my own.

 $^{^{35}}$ The BLS defines long-term unemployment as being jobless for 27 weeks or longer.

³⁶ The IJS effects are also similar by race (Black vs. non-Black) or ethnicity (Hispanic vs. non-Hispanic).

Appendix

Table A1

Longitudinally Linked CPS Files.

	Month 0	Month 1	Month 2	Month 3	Month 9	Month 10	Month 11	Month 12	Month 13	Month 14	Month 15
	1983/09	1983/10	1983/11	1983/12	1984/06	1984/07	1984/08	1984/09	1984/10	1984/11	1984/12
	1989/12	1990/01	1990/02	1990/03	1990/09	1990/10	1990/11	1990/12	1991/01	1991/02	1991/03
	1991/08	1991/09	1991/10	1991/11	1992/05	1992/06	1992/07	1992/08	1992/09	1992/10	1992/11
	1992/09	1992/10	1992/11	1992/12	1993/06	1993/07	1993/08	1993/09	1993/10	1993/11	1993/12
	1998/12	1999/01	1999/02	1999/03	1999/09	1999/10	1999/11	1999/12	2000/01	2000/02	2000/03
	2000/08	2000/09	2000/10	2000/11	2001/05	2001/06	2001/07	2001/08	2001/09	2001/10	2001/11
	2001/09	2001/10	2001/11	2001/12	2002/06	2002/07	2002/08	2002/09	2002/10	2002/11	2002/12
	2003/10	2003/11	2003/12	2004/01	2004/07	2004/08	2004/09	2004/10	2004/11	2004/12	2005/01
	2011/07	2011/08	2011/09	2011/10	2012/04	2012/05	2012/06	2012/07	2012/08	2012/09	2012/10
1	2013/07	2013/08	2013/09	2013/10	2014/04	2014/05	2014/06	2014/07	2014/08	2014/09	2014/10
	2015/07	2015/08	2015/09	2015/10	2016/04	2016/05	2016/06	2016/07	2016/08	2016/09	2016/10

Used for main analysis

Used for placebo exercise

Added for construction of extended sample

Table A2

Pre-Internet Computer Use at Work, by Occupation.

Category Number	CPS detailed occupation	Proportion using computers at work (C93)	
1	Administrators and officials, public administration	0.862	
2	Other executive, administrators, and managers	0.882	
3	Management related occupations	0.872	
4	Engineers	0.895	
5	Mathematical and computer scientists	0.984	
6	Natural scientists	0.873	
7	Health diagnosing occupations	0.490	
8	Health assessment and treating occupations	0.596	
9	Teachers, college and university	0.725	
10	Teachers, except college and university	0.490	
10	Lawyers and judges	0.646	
12	Other professional specialty occupations	0.581	
13	Health technologists and technicians	0.554	
14	Engineering and science technicians	0.691	
15	Technicians, except health, engineering, and science	0.897	
16	Supervisors and proprietors, sales occupations	0.571	
17	Sales representatives, finance, and business service	0.732	
18	Sales representatives, commodities, except retail	0.560	
19	Sales workers, retail and personal services	0.298	
20	Sales related occupations	0.233	
20 21	Supervisors - administrative support	0.868	
22	Computer equipment operators	0.965	
22	Secretaries, stenographers, and typists	0.862	
23	Financial records, processing occupations	0.882	
25	Mail and message distributing	0.259	
25	Other administrative support occupations, including clerical	0.744	
20 27	Private household service occupations	0.006	
27 28	Protective service occupations	0.000	
28	Food service occupations	0.108	
30	Health service occupations	0.108	
31	Cleaning and building service occupations	0.053	
31	Personal service occupations	0.053	
33	Mechanics and repairers	0.102	
34	Construction trades	0.079	
35 36	Other precision production occupations	0.347	
36	Machine operators and tenders, except precision	0.175 0.175	
38	Fabricators, assemblers, inspectors, and samplers	0.175	
38	Motor vehicle operators	0.121	
	Other transportation occupations and material moving		
40	Construction laborer	0.036	
41	Freight, stock and material handlers	0.158	
42	Other handlers, equipment cleaners, and laborers	0.126	
43	Farm operators and managers	0.137	
44	Farm workers and related occupations	0.059	
45	Forestry and fishing occupations	0.029	
46	Armed forces last job, currently unemployed	0.452	
N/A	Never worked	0.000	

Notes. Data are from the October 1993 CPS School Enrollment Supplement. The detailed occupations based on the 1990 Census occupational classification are used. For those who served in the armed forces just before becoming unemployed, the computer use rate at work is imputed with the average across all 45 occupations (i.e., 0.452). For those who never had a job, the computer use rate at work is set to zero.

Table A3

IV Estimates of Internet Job Search Effects on Employment Probability, by Various IV Specifications.

Dependent variable:	Employed within	Employed in			
	15 months (1)	month 1 (2)	month 2 (3)	month 12 (4)	
A. Linear time trend: $C93_k \times t$					
IJS	0.266**	0.255***	0.177	0.003	
	(0.112)	(0.087)	(0.125)	(0.162)	
First-stage F-statistic	39.93	35.60	27.67	32.61	
B. Quadratic time trend: $C93_k \times$	f(t)				
IIS	0.160**	0.163***	0.126*	0.110	
	(0.069)	(0.059)	(0.076)	(0.105)	
First-stage F-statistic	67.65	62.60	66.16	46.82	
C. Cubic time trend: $C93_k \times f(t)$					
IJS	0.148**	0.158***	0.116	0.068	
	(0.068)	(0.059)	(0.076)	(0.100)	
First-stage F-statistic	53.61	49.67	47.21	37.58	
D. Time dummies: $C93_k \times \lambda_t$					
IJS	0.129**	0.130**	0.105	0.043	
5	(0.065)	(0.056)	(0.075)	(0.093)	
First-stage F-statistic	37.07	32.62	32.86	28.12	
Ν	15,655	13,613	8,679	6,301	
Mean of Y	0.514	0.265	0.344	0.524	

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Besides occupation and year fixed effects, control variables include unemployment duration (in weeks), state and occupational unemployment rates, and dummies for age (16–25, 26–35, 36–45, 46–55), education (some high school, high school graduate, some college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. The first-stage *F*-statistics are computed by implementing the weak instrument test by Montiel Olea and Pflueger (2013). Robust standard errors in parentheses are clustered at the occupation-by-year level. * p < 0.00, ** p < 0.05, *** p < 0.01.

Table A4

Reduced-Form Estimates on Employment Probability, by Alternative Treatments of the Never-Worked.

Dependent variable:	Employed within 15 months						
	Main sample (1)	Impute with $C93 = 0.053$ (2)	Impute with C93 = 0.108 (3)	Impute with $C93 = 0.175$ (4)	Without the never-worked (5)		
A. IV: $C93_k \times \lambda_t$							
$C93 \times 1[t = 1998]$	0.120***	0.118***	0.116***	0.112***	0.096**		
	(0.040)	(0.040)	(0.040)	(0.040)	(0.040)		
$C93 \times 1[t = 2000]$	0.097**	0.092**	0.086**	0.077*	0.048		
	(0.041)	(0.041)	(0.041)	(0.040)	(0.037)		
$C93 \times 1[t = 2001]$	0.072*	0.067*	0.061*	0.053	0.028		
	(0.037)	(0.037)	(0.036)	(0.035)	(0.033)		
$C93 \times 1[t = 2003]$	0.048	0.045	0.040	0.033	0.016		
	(0.033)	(0.033)	(0.032)	(0.032)	(0.030)		
$C93 \times 1[t = 2011]$	0.120***	0.118***	0.114***	0.109***	0.092**		
	(0.042)	(0.042)	(0.042)	(0.042)	(0.042)		
F-statistic for joint	2.881	2.743	2.609	2.436	1.849		
significance test	[0.015]	[0.019]	[0.025]	[0.035]	[0.103]		
Adjusted R ²	0.079	0.079	0.079	0.079	0.081		
B. IV: C93 _k × t							
$C93 \times t$	0.005**	0.005**	0.005**	0.005**	0.004*		
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)		
Adjusted R ²	0.079	0.079	0.079	0.079	0.081		
Ν	15,655	15,655	15,655	15,655	14,534		
Mean of Y	0.514	0.514	0.514	0.514	0.522		

Notes. Data are from the September 1992 Basic Monthly CPS and the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements, matched with their subsequent monthly CPS files. C93 denotes the 1993 computer use rate at work in each occupation, calculated using data from the October 1993 CPS School Enrollment Supplement. Column (1) reproduces the reduced-form estimates in Table 4. In columns (2)–(4), C93 for those without work experience is imputed with the value of C93 in occupation categories 31, 29, and 36, respectively. In column (5), the estimation sample excludes individuals who never worked before the current unemployment spell. Besides occupation and year fixed effects, control variables include unemployment function (in weeks), state and occupational unemployment rates, and dumnies for age (16–25, 26–35, 36–45, 46–55), education (some high school graduate, some college, college graduate), Black, Hispanic, male, married, married male, spousal employment, home ownership, job quitters, on layoff, status before unemployment (worked, attended school), sector (private, public, self-employed), industry, and state. Robust standard errors in parentheses are clustered at the occupation-by-year level. *p*-values in brackets. * *p* < 0.10, *** *p* < 0.05, *** *p* < 0.01.



Fig. A1. Proportion of Unemployed Workers Searching for Jobs Online, by Occupation (All Occupations). Notes. Online job search status is from the December 1998, August 2000, September 2001, October 2003, and July 2011 CPS Computer and Internet Use Supplements. C93 denotes the 1993 computer use rate at work in each occupation from the October 1993 CPS School Enrollment Supplement. Occupation codes follow the 1990 Census occupational classification.

Supplementary material

Supplementary material associated with this article can be found, in the online version, at doi:10.1016/j.infoecopol.2023. 101017

References

- Abadie, A., Athey, S., Imbens, G.W., Wooldridge, J., 2017. When Should You Adjust Standard Errors for Clustering? Working Paper 24003. National Bureau of Economic Research doi:10.3386/w24003.
- Abraham, K.G., Spletzer, J.R., 2010. Are the new jobs good jobs? In: Abraham, K.G., Spletzer, J.R., Harper, M. (Eds.) Labor in the New Economy. University of Chicago Press, pp. 101–143. chapter 3, https://www.nber.org/ books-and-chapters/labor-new-economy/are-new-jobs-good-jobs.
- Acemoglu, D., Johnson, S., 2007. Disease and development: the effect of life expectancy on economic growth. J. Polit. Economy 115 (6), 925–985. doi:10.1086/ 529000.
- Angrist, J.D., 2001. Estimation of limited dependent variable models with dummy endogenous regressors. J. Bus. Econ. Stat. 19 (1), 2–28. doi:10.1198/ 07350010152472571.
- Autor, D.H., 2001. Wiring the labor market. J. Econ. Perspect. 15 (1), 25–40. doi:10. 1257/jep.15.1.25.
- Autor, D.H., Katz, L.F., Krueger, A.B., 1998. Computing inequality: have computers changed the labor market? Q. J. Econ. 113 (4), 1169–1213. doi:10.1162/ 003355398555874.
- Bagues, M.F., Labini, M.S., 2009. Do online labor market intermediaries matter? The impact of AlmaLaurea on the university-to-work transition. In: Autor, D.H. (Ed.), Studies of Labor Market Intermediation. The University of Chicago Press, pp. 127–154. chapter 4, https://www.nber.org/booksand-chapters/studies-labor-market-intermediation/do-online-labor-marketintermediaries-matter-impact-almalaurea-university-work-transition.

- Bellou, A., 2015. The impact of internet diffusion on marriage rates: evidence from the broadband market. J. Popul. Econ. 28 (2), 265–297. doi:10.1007/ s00148-014-0527-7.
- Bhuller, M., Kostøl, A., Vigtel, T., 2021. The Internet, Search Frictions and Aggregate Unemployment. Working Paper. https://www.dropbox.com/s/gc5bkco6xfwlp1m/ BKV_Jan2021.pdf
- Brown, J.R., Goolsbee, A., 2002. Does the internet make markets more competitive? Evidence from the life insurance industry. J. Polit. Economy 110 (3), 481–507. http://www.jstor.org/stable/10.1086/339714
- Cameron, C.A., Trivedi, P.K., 2005. Microeconometrics: Methods and Applications. Cambridge University Press. https://www.google.com/books/edition/_/ TdlKAgAAQBAJ?hl=en&gbpv=0.
- Cumming, C.S., Bahn, K., Zickuhr, K., 2022. How New Job Search Technologies are Affecting the U.S. Labor Market. Washington Center for Equitable Growth. https://equitablegrowth.org/how-new-job-search-technologiesare-affecting-the-u-s-labor-market/.
- Denzer, M., Schank, T., Upward, R., 2021. Does the internet increase the job finding rate? Evidence from a period of expansion in internet use. Inf. Econ. Policy 55, 100900. doi:10.1016/j.infoecopol.2020.100900.
- Duflo, E., 2001. Schooling and labor market consequences of school construction in indonesia: evidence from an unusual policy experiment. Am. Econ. Rev. 91 (4), 795–813. doi:10.1257/aer.91.4.795.
- Fountain, C., 2005. Finding a job in the internet age. Social Forces 83 (3), 1235–1262. doi:10.1353/sof.2005.0030.
- Gürtzgen, N., Diegmann, A., Pohlan, L., van den Berg, G.J., 2021. Do digital information technologies help unemployed job seekers find a job? Evidence from the broadband internet expansion in germany. Eur. Econ. Rev. 132, 103657. doi:10.1016/j.euroecorev.2021.103657.
- Kroft, K., Pope, D.G., 2014. Does online search crowd out traditional search and improve matching efficiency? Evidence from Craigslist. J. Labor Econ. 32 (2), 259– 303. doi:10.1086/673374.
- Krueger, A.B., 1993. How computers have changed the wage structure: evidence from microdata, 1984–1989. Q. J. Econ. 108 (1), 33–60. doi:10.2307/2118494.
- Kuhn, P., Mansour, H., 2014. Is internet job search still ineffective? Econ. J. 124 (581), 1213–1233. doi:10.1111/ecoj.12119.

- Kuhn, P., Skuterud, M., 2004. Internet job search and unemployment durations. Am. Econ. Rev. 94 (1), 218–232. doi:10.1257/000282804322970779.
- Madrian, B.C., Lefgren, L.J., 2000. An approach to longitudinally matching current population survey (CPS) respondents. J. Econ. Soc. Meas. 26 (1), 31–62. doi:10. 3233/JEM-2000-0165.
- Montiel Olea, J.L., Pflueger, C., 2013. A robust test for weak instruments. J. Bus. Econ. Stat. 31 (3), 358–369. doi:10.1080/00401706.2013.806694.
- Mortensen, D.T., Pissarides, C.A., 1999. New developments in models of search in the labor market. In: Ashenfelter, O.C., Card, D. (Eds.), Handbook of Labor Economics, Vol. 3, Part 2. Elsevier, pp. 2567–2627. doi:10.1016/S1573-4463(99) 30025-0. chapter 36
- Nakamura, A.O., Shaw, K.L., Freeman, R.B., Nakamura, E., Pyman, A., 2009. Jobs online. In: Autor, D.H. (Ed.), Studies of Labor Market Intermediation. The University of Chicago Press, pp. 27–65. chapter 1, https://www.nber.org/ books-and-chapters/studies-labor-market-intermediation/jobs-online/.
- Pew Research Center, 2015. Searching for work in the digital era. https://www. pewresearch.org/internet/2015/11/19/searching-for-work-in-the-digital-era/.
- Pflueger, C.E., Wang, S., 2015. A robust test for weak instruments in stata. Stata J. 15 (1), 216–225. doi:10.1177/1536867X1501500113.

- Pissarides, C.A., 2000. Equilibrium Unemployment Theory, second ed. The MIT Press. https://mitpress.mit.edu/9780262533980/equilibrium-unemployment-theory/.
- Stevenson, B., 2009. The internet and job search. In: Autor, D.H. (Ed.), Studies of Labor Market Intermediation. The University of Chicago Press, pp. 67–86. doi:10. 3386/w13886. chapter 2
- Stock, J.H., Yogo, M., 2005. Testing for weak instruments in linear IV regression. In: Andrews, D.W.K., Stock, J.H. (Eds.), Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg. Cambridge University Press. https://scholar.harvard.edu/stock/publications/ testing-weak-instruments-linear-iv-regression.
- Van den Berg, G.J., 1994. The effects of changes of the job offer arrival rate on the duration of unemployment. J. Labor Econ. 12 (3), 478–498. doi:10.1086/298353.
- Van den Berg, G.J., Van der Klaauw, B., 2006. Counseling and monitoring of unemployed workers: theory and evidence from a controlled social experiment. Int. Econ. Rev. 47 (3), 895–936. doi:10.1111/j.1468-2354.2006.00399.x.
- Zhao, D., 2019. The rise of mobile devices in job search: challenges and opportunities for employers. https://www.glassdoor.com/research/mobile-job-search/